Author's response to reviews

Title: Effect of a Proprietary Magnolia and Phellodendron Extract on Stress Levels in Healthy Women: a pilot, double-blind, placebo-controlled clinical trial

Authors:

Douglas S Kalman (dkalman@miamiresearch.com)
Samantha Feldman (sfeldman@miamiresearch.com)
Robert Feldman (rfeldman@miamiresearch.com)
Howard I Schwartz (hschwartz@miamiresearch.com)
Diane R Krieger (drkrieger@miamiresearch.com)
Robert Garrison (ddolnick@nextpharmaceuticals.com)

Version: 3 Date: 18 April 2008

Author's response to reviews: see over
April 18, 2008

Nutrition Journal

Thank you for accepting our study “Effect of a proprietary Magnolia and Phellodendron extract on stress levels in healthy women: a pilot, double-blind, placebo-controlled clinical trial”.

We have made the formatted changes as requested.

- The title was changed to sentence case, all unnecessary capitals were replaced by lower case letters.
- Author qualifications were removed.
- Affiliations were corrected and the country was added to each one.
- Principal Investigator/ Sub-Investigators’ information was removed from the title page.
- The references were corrected, removed “et al.” & added authors that were missing.
- Table order – was corrected and the mention of Table 1 was included.

Please note that all authors have read and approved the final manuscript.

Sincerely,

Douglas Kalman PhD, RD
Miami Research Associates.